Author's response to reviews

Title: Do we need more than one Child Perceptions Questionnaire for children and adolescents?

Authors:

Lyndie A Foster Page (lyndie.fosterpage@otago.ac.nz)
Dorothy H Boyd (dorothy.boyd@otago.ac.nz)
W Murray Thomson (murray.thomson@otago.ac.nz)

Version: 3 Date: 28 May 2013

Author's response to reviews: see over
May 28th 2013

Response to editorial board member.

Thank you for your time in reading this paper and your comments

We have highlighted our changes in blue.

1. We have amended the paper to account for the older and younger children as suggested in the analysis and have analysed the construct validity by the two different age groups and the correlational validity.
2. We have removed the term valid in the conclusion and discussion and discussed the language issues and the need to validate this for this age group.
3. We have stated that this study is the first stage in a process of validating this questionnaire in the 5-8 yr old age group and the need for further research with this age group.